Author’s response to reviews

Title: Ultrasound-guided identification of the cricothyroid membrane in a patient with a difficult airway: A case report

Authors:

Hiromu Okano (okanohiromu0121@gmail.com)

Kohji Uzawa (kohji.fentanyl@gmail.com)

Kunitaro Watanabe (kunitarowatanabe@yahoo.co.jp)

Akira Motoyasu (akira.aka.akrstyle@gmail.com)

Joho Tokumine (ii36469@wa2.so-net.ne.jp)

Alan Lefor (alefor@jichi.ac.jp)

Tomoko Yorozu (tyorozu@gmail.com)

Version: 2 Date: 24 Jan 2018

Author’s response to reviews:

1. Can you please clarify if you obtained consent from the patient for publication of all images? If yes, then please clearly state this in the ‘Consent for publication’ section.

Thank you very much for your suggestion. I described it below.

Written informed consent for publication of the clinical details and clinical images was obtained from the patient.

2. Thank you for the authors’ contribution section. Unfortunately, not all of the authors have made significant contributions to the study.

Thank you very much for your suggestion. I described it in detail.

Authors' contributions:

HO: Acquisition of data or analysis and interpretation of data

KU: writing manuscript (discussion and conclusions) and searching literature
KW: writing manuscript (case presentation), and conception and design

AM: writing manuscript (discussion and conclusions)

JT: Conception and design, acquisition of data or analysis and interpretation of data

AKL: Drafting the article or revising it critically for important intellectual content.

TY: Final approval of the version published.

3. Please make sure that in the authors contributions section all authors are listed by their initials not their full names.

Thank you very much for your suggestion. I rewrote them as above.

4. On uploading your revisions, please remove any tracked changes or highlighting and include only a single clean copy of the manuscript.

I will resubmit it our revision as you suggested.